

September 3, 2014

Dear Editor,



Please find enclosed the edited manuscript in Word format (file name: 13041-edited.docx).

Title: Safety of biologic therapies during pregnancy in women with rheumatic disease

Author: Natalia Mena-Vazquez, Sara Manrique-Arija, Antonio Fernandez-Nebro

Name of Journal: *World Journal of Rheumatology*

ESPS Manuscript NO: 13041

The linguistic errors reported by the reviewer have been solved.

Paragraph "Author institution" has been added.

Thank you again for publishing our manuscript in the *World Journal of Rheumatology*.

Sincerely yours,

A handwritten signature in black ink, appearing to be 'Antonio Fernández-Nebro', written over a circular stamp or watermark.

Antonio Fernández-Nebro, PhD

UGC Reumatología, Instituto de Investigación Biomédica de Málaga (IBIMA), Hospital Regional Universitario de Málaga, Universidad de Málaga

Plaza del Hospital Civil s/n

Pabellón 7-2ª planta

29009 - Málaga, Spain

aternandezn@uma.es

Tel/Fax number: +34951290360